Author's response to reviews

Title: Developing Family Rooms in Mental Health inpatient units: an exploratory descriptive study.

Authors:

Sophie Isobel (sophie.norrish@sswahs.nsw.gov.au)
Kim Foster (kim.foster@canberra.edu.au)
Clair Edwards (clair.edwards@sswahs.nsw.gov.au)

Version: 4  Date: 14 May 2015

Author's response to reviews:

Dear Editor,

As per your requests the manuscript has now had the following changes and is ready for publication.

1. The name of the Ethics Committee is included
2. There are no abbreviations but this is now noted
3. Tables have been removed from figure files
4. Correct headings have been used.
5. All changes are highlighted in the amended manuscript.

Thank you for the opportunity to publish in your journal.

Kind Regards,

Sophie Isobel